Author's response to reviews

Title: Real-life Effectiveness of Budesonide/Formoterol Maintenance and Reliever Therapy in Asthma Patients Across Asia: SMARTASIA study

Authors:

Nanshan Zhong (nanshan@vip.163.com)
JiangTao Lin (jiangtao_l@263.net)
Parthiv Mehta (parthiv@parthivs.com)
Pintip Ngamjanyaporn (pintiptip@yahoo.com)
Tzu-Chin Wu (tcwu@csmu.edu.tw)
Faisal Yunus (faisalyunus@ymail.com)
Jing-Ping Zheng (jpzhenggy@163.com)

Version: 3 Date: 21 December 2012

Author's response to reviews: see over
Dear Editor,

Revision of Article (MS: 4745778027722560)

Thank you for your kind review of our manuscript, “Real-life Effectiveness of Budesonide/Formoterol Maintenance and Reliever Therapy in Asthma Patients Across Asia: SMARTASIA study”.

We thank the two excellent reviewers for their meticulous review of our manuscript, valuable comments and suggestions which have enabled us to come out with an improved, much better version of the manuscript.

We have addressed all issues raised by the reviewers, including supplementing existing information with results derived from additional data sub-analysis to further support our report of the Asia-wide observational study. The changes in the manuscript are in blue fonts.

A point-to-point response to the reviewers’ queries is attached for your kind reference of all the amendments made in this version of the manuscript, and the rationale behind the statements made in the manuscript.

We look forward to your kind consideration of the manuscript’s publication in your most esteemed journal, BMC Pulmonary Medicine.

Thank you.

Yours truly,

Nanshan Zhong

Professor,
Respiratory Research Institute
1st Affiliated Hospital of Guangzhou Medical College
Guangzhou, China

Email: nanshan@vip.163.com